**Author's response to reviews**

**Title:** Host factors do not influence the colonization or infection by fluconazole resistant Candida species in hospitalized patients

**Authors:**

Yun-Liang Yang (yyang@mail.nctu.edu.tw)
Ming-Fang Cheng (mfcheng@vghks.gov.tw)
Ya-Wen Chang (debbychang@nhri.org.tw)
Tzuu-Guang Young (dal07@tpech.gov.tw)
Hsin Chi (a4531@ttms.mmh.org.tw)
Sai Cheong Lee (leesc@url.com.tw)
Bruno Man-Hon Cheung (mhcheung2002@yahoo.com.tw)
Fan-Chen Tseng (950119@nhri.org.tw)
Tun-Chieh Chen (880305@ms.kmuh.org.tw)
Yu-Huai Ho (hoyuhuai@yahoo.com.tw)
Zhi-Yuan Shi (a4531@ttms.mmh.org.tw)
Chung-Huang Hubert Chan (hubertchan@nhri.org.tw)
Ju-Yu Lin (m508092009@tmu.edu.tw)
Hsiu-Jung Lo (hjlo@nhri.org.tw)

**Version:** 3  **Date:** 11 December 2008

**Author's response to reviews:**

To whom it may concern,

Thank you for publishing our work entitled “Host factors do not influence the colonization or infection by fluconazole resistant Candida species in hospitalized patients”.

We have added available email addresses all authors in the title page.

We have reformatted the references in the order that they are listed in the references section, and that they each follow on numerically.

We have revised tables as suggested.

Hsiu-Jung Lo